Author’s response to reviews

Title: Cesarean scar endometriosis: Presentation of 198 cases and literature review

Authors:

Ping Zhang (shping1216@163.com)
Yabing Sun (yabingsun1990@126.com)
Chen Zhang (13916681426@163.com)
Yeping Yang (yyp0250@126.com)
Linna Zhang (1099023229@qq.com)
Ningling Wang (nlingwang@163.com)
Hong Xu (xuhong1168@126.com)

Version: 2 Date: 28 Dec 2018

Author’s response to reviews:

Dear editor,

Thank you for your letter and the comments concerning our manuscript entitled “Cesarean scar endometriosis: Presentation of 198 cases and literature review” (BMWH-D-18-00336R1). We read the comments carefully and made corrections according to the comments. Our point-by-point response to each comment is shown below.

1. Author’s emails: Please include in the title page the email addresses of all authors. The corresponding author should still be indicated. Please ensure these email addresses match the email addresses provided in the editorial manager system.

Response: Email addresses of all authors have been included in the title page. (Page 1)

2. Headings and formatting: In order to be in line with journal requirements, please use the following headings within your manuscript: Background, Methods, Results, Discussion and Conclusion.

Response: The revised manuscript conforms to the journal style.
3. Funding: The role of the funding body in the design of the study and collection, analysis and interpretation of data and in writing the manuscript should be declared.

Response: The role of the funding body has been declared in the revised manuscript. (Page 14)

4. Clean manuscript Final Checks: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Response: Unnecessary highlights and colours have been removed.

We appreciate the editors’ and reviewers’ kind work earnestly, and we hope that the corrections are sufficient to qualify for approval. Once again, thank you very much for your comments and suggestions.

Sincerely yours,